Author's response to reviews

Title: Pseudoaneurysm of graft-graft anastomosis of a hand-sewn branched graft

Authors:

Toshihito Gomibuchi (gomit@shinshu-u.ac.jp)
Tamaki Takano (ttakano-ths@umin.ac.jp)
Yuko Wada (aruko@shinshu-u.ac.jp)
Takamitsu Terasaki (mayflyt@shinshu-u.ac.jp)
Tatsuichiro Seto (seto@shinshu-u.ac.jp)
Daisuke Fukui (dfukui@shinshu-u.ac.jp)

Version: 2
Date: 30 March 2015

Author's response to reviews: see over
Mr. Vipin Zamvar  
Editor-in-Chief, Journal of Cardiothoracic Surgery

Dear Mr. Vipin:
On behalf of all the authors, I would like to ask you to consider our manuscript entitled "Pseudoaneurysm of graft–graft anastomosis of a hand-sewn branched graft" for publication in Journal of Cardiothoracic Surgery as a case report. This report describes a very rare case of a pseudoaneurysm at the graft–graft anastomosis of a hand-sewn branched graft 22 years after total arch replacement, which is of special interest because this kind of report has not been previously published. Patient in this report have provided permission to publish these features of their cases, and the identity of the patients has been protected.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. The authors report no conflicts of interest.
Thank you for your time

Sincerely,

Gomibuchi Toshihito M.D.
Department of Cardiovascular Surgery, Shinshu University School of Medicine
3-3-1 Asahi, Matsumoto
Nagano, JAPAN

Tel: +81-263-37-2657
Fax: +81-263-37-2721
gomit@shinshu-u.ac.jp